Author's response to reviews

Title: Use of a Virtual World Computer Environment for International Distance Education: Lessons from a Pilot Project using Second Life

Authors:

Marloes Schoonheim (marloes161@gmail.com)
Robin Heyden (robinheyden@gmail.com)
John Wiecha (john383@bu.edu)

Version: 2 Date: 11 October 2013

Author's response to reviews:

The revisions include bringing the paper into compliance with guidelines for a Correspondence category, including word count, and section headers.